\begin{thebibliography}{BMPT90}

\bibitem[Acz77]{Acz77}
Peter Aczel.
\newblock An introduction to inductive definitions.
\newblock In J.~Barwise, editor, {\em Handbook of Mathematical Logic},
  volume~90 of {\em Studies in Logic and the Foundations of Mathematics},
  chapter C.7, pages 739--782. North-Holland, Amsterdam, 1977.

\bibitem[Apt90]{Apt90}
Krzysztof~R. Apt.
\newblock Logic programming.
\newblock In J.~van Leewen, editor, {\em Handbook of Theoretical Computer
  Science}, volume B: Formal Models and Semantics, chapter~10, pages 493--574.
  The MIT Press, New York, N.Y., 1990.

\bibitem[BM90]{Bonner90}
Anthony~J. Bonner and L.~Thorne McCarty.
\newblock Adding negation-as-failure to intuitionistic logic programming.
\newblock In Saumya~Hermenegildo Manuel~Debray, editor, {\em Proceedings of the
  1990 North American Conference on Logic Programming}, pages 671--693, Austin,
  TX, October 1990. MIT Press.

\bibitem[BMPT90]{Bar90}
Roberto Barbuti, Paolo Mancarella, Dino Pedreschi, and Franco Turini.
\newblock {A} transformational approach to negation in logic programming.
\newblock {\em Journal of Logic Programming}, 8:201--228, 1990.

\bibitem[BMV89]{Bonner89}
A.~J. Bonner, L.~T. McCarty, and K.~Vadaparty.
\newblock Expressing {D}atabase {Q}ueries with {I}ntuitionistic {L}ogic.
\newblock In Ewing~L. Lusk and Ross~A. Overbeek, editors, {\em Proceedings of
  the North American Conference on Logic Programming}, pages 831--850,
  Cleveland, Ohio, USA, 1989.

\bibitem[Bon91]{Bonner91phd}
A.J. Bonner.
\newblock {\em Hypothetical Reasoning in Deductive Databases}.
\newblock PhD thesis, Rutgers University, October 1991.

\bibitem[Bon94]{Bonner94}
A.J. Bonner.
\newblock Hypothetical reasoning with intuitionistic logic.
\newblock In R.~Demolombe and T.~Imielinski, editors, {\em Non-Standard Queries
  and Answers}, volume 306 of {\em Studies in Logic and Computation}, pages
  187--219. Oxford University Press, 1994.

\bibitem[Cla78]{Cla78}
K.~L. Clark.
\newblock Negation as failure.
\newblock In H.~Gallaire and J.~Minker, editors, {\em Logic and Databases},
  pages 293--322. Plenum Press, New York, 1978.

\bibitem[Dun92]{Dung92}
P.~M. Dung.
\newblock Declarative semantics of hypothetical logic programming with negation
  as failure.
\newblock In E.~Lamma and P.~Mello, editors, {\em Proceedings of the Third
  International Workshop on Extensions of Logic Programming ({ELP} '92)},
  volume 660 of {\em LNAI}, pages 45--58, Bologna, Italy, February 1992.
  Springer Verlag.

\bibitem[Eri93]{Eriksson93phd}
Lars-Henrik Eriksson.
\newblock {\em Finitary Partial Inductive Definitions and General Logic}.
\newblock PhD thesis, Department of Computer and System Sciences, Royal
  Institute of Technology, Stockholm, 1993.

\bibitem[Gab85]{Gabbay85}
Dov~M. Gabbay.
\newblock N-{P}rolog: An extension of prolog with hypothetical implications
  {II}. logical foundations and negation as failure.
\newblock {\em Journal of Logic Programming}, 2(4):251--283, December 1985.

\bibitem[GL90]{Gelfond90}
Michael Gelfond and Vladimir Lifschitz.
\newblock Logic programs with classical negation.
\newblock In David H.D.~Szerdei Peter~Warren, editor, {\em Proceedings of the
  7th International Conference on Logic Programming (ICLP '90)}, pages
  579--597, Jerusalem, June 1990. MIT Press.

\bibitem[GMR92]{Giordano92}
L.~Giordano, A.~Martelli, and G.~Rossi.
\newblock Extending {Horn} clause logic with implication goals.
\newblock {\em Theoretical Computer Science}, 95(1):43--74, March 1992.

\bibitem[GO98]{Giordano98}
Laura Giordano and Nicola Olivetti.
\newblock Negation as failure and embedded implication.
\newblock {\em Journal of Logic Programming}, to appear, 1998.

\bibitem[GR84]{Gabbay84}
Dov~M. Gabbay and U.~Reyle.
\newblock N-{P}rolog: An extension of prolog with hypothetical implications
  {I}.
\newblock {\em Journal of Logic Programming}, 1(4):319--355, December 1984.

\bibitem[Hal91]{Halnass91}
L.~Hallnas.
\newblock Partial inductive definitions.
\newblock {\em Theoretical Computer Science}, 87(1):115--147, July 1991.

\bibitem[Har89]{Harland89}
J.~Harland.
\newblock A {K}ripke-like {M}odel for {N}egation as {F}ailure.
\newblock In Ewing~L. Lusk and Ross~A. Overbeek, editors, {\em Proceedings of
  the North American Conference on Logic Programming}, pages 626--644,
  Cleveland, Ohio, USA, 1989.

\bibitem[Har91]{Harland91phd}
James Harland.
\newblock {\em On Hereditary Harrop Formulae as a Basis for Logic Programming}.
\newblock PhD thesis, Edinburgh, January 1991.

\bibitem[Har93]{Harland93}
James Harland.
\newblock Success and failure for hereditary {Harrop} formulae.
\newblock {\em Journal of Logic Programming}, 17(1):1--29, October 1993.

\bibitem[HM94]{Hodas94ic}
Joshua Hodas and Dale Miller.
\newblock Logic programming in a fragment of intuitionistic linear logic.
\newblock {\em Information and Computation}, 110(2):327--365, 1994.
\newblock A preliminary version appeared in the Proceedings of the Sixth Annual
  IEEE Symposium on Logic in Computer Science, pages 32--42, Amsterdam, The
  Netherlands, July 1991.

\bibitem[HSH90]{Halnass90}
L.~Hallnas and P.~Schroeder-Heister.
\newblock A proof-theoretic approach to logic programming: Clauses as rules.
\newblock {\em Journal of Logic and Computation}, 1(2):635\--660, October 1990.

\bibitem[HSH91]{Halnass91jlp}
L.~Hallnas and P.~Schroeder-Heister.
\newblock A proof-theoretic approach to logic programming: Programs as
  definitions.
\newblock {\em Journal of Logic and Computation}, 1(5):261\--283, October 1991.

\bibitem[KNW93]{Kwon93}
Keehand Kwon, Gopalan Nadathur, and Debra~Sue Wilson.
\newblock Implementing a notion of modules in the logic programming language
  {$\lambda$Prolog}.
\newblock In E.~Lamma and P.~Mello, editors, {\em Proceedings of the Third
  International Workshop on Extensions of Logic Programming}, pages 359--393,
  Bologna, Italy, February 1993. Springer-Verlag LNAI 660.

\bibitem[Kre92]{Kreuger92}
Per Kreuger.
\newblock {GCLA} {II}. a definitional approach to control.
\newblock Technical Report R92:09, SICS, 1992.

\bibitem[MAK91]{Aronson91}
L.~Hallnass Martin~Aronson, L-H.~Eriksson and P.~Kreuger.
\newblock A survey of {GCLA}: a definitional approach to logic programming.
\newblock In Peter Schroeder-Heister, editor, {\em Proceedings of the First
  International Workshop on Extensions of Logic Programming}, volume 1050 of
  {\em 475}, pages 19--34, Berlin, March28--30~ 1991. Springer.

\bibitem[McC88a]{Mc88a}
L.~Thorne McCarty.
\newblock Clausal intuistionistic logic: {I}. fixed point semantics.
\newblock {\em Journal of Logic Programming}, 5(1):1--31, March 1988.

\bibitem[McC88b]{Mc88b}
L.~Thorne McCarty.
\newblock Clausal intuitionistic logic {II}. tableau proof procedure.
\newblock {\em Journal of Logic Programming}, 5:93--132, 1988.

\bibitem[Mil89]{Miller89jlp}
Dale Miller.
\newblock A logical analysis of modules in logic programming.
\newblock {\em Journal of Logic Programming}, 6(1-2):79--108, January 1989.

\bibitem[MM97]{MacDowell97lics}
Raymond McDowell and Dale Miller.
\newblock A logic for reasoning with higher-order abstract syntax: An extended
  abstract.
\newblock In Glynn Winskel, editor, {\em Proceedings of the Twelfth Annual
  Symposium on Logic in Computer Science}, pages 434--445, Warsaw, Poland, June
  1997.

\bibitem[MMP88]{Mancarella88}
Paolo Mancarella, Simone Martini, and Dino Pedreschi.
\newblock {C}omplete logic {P}rograms with domain-closure {A}xiom.
\newblock {\em Journal of Logic Programming}, 5:263--276, 1988.

\bibitem[Mom92]{Momigliano92}
Alberto Momigliano.
\newblock Minimal negation and hereditary {H}arrop formulae.
\newblock In Anil Nerode and Mikhail Taitslin, editors, {\em Proceedings of
  Logical Foundations of Computer Science (Tver '92)}, volume 620 of {\em
  LNCS}, pages 326--335, Berlin, Germany, July 1992. Springer.

\bibitem[Nel49]{Nelson49}
D.~Nelson.
\newblock Constructive falsity.
\newblock {\em Journal of Symbolic Logic}, pages 16--26, 1949.

\bibitem[NL92]{Olivetti92}
Olivetti N. and Terracini L.
\newblock N-{P}rolog and the equivalence of logic program.
\newblock {\em Journal of Logic Language and Information}, 5:253\--3392, 1992.

\bibitem[NL95]{Nadathur95lics}
Gopalan Nadathur and Donald~W. Loveland.
\newblock Uniform proofs and disjunctive logic programming.
\newblock In D.~Kozen, editor, {\em Proceedings of the Tenth Annual Symposium
  on Logic in Computer Science}, pages 148--155, San Diego, California, June
  1995. IEEE Computer Society Press.
\newblock Available as Technical Report CS-1994-40, Department of Computer
  Science, Duke University, December 1994.

\bibitem[Pea90]{Pearce?}
David Pearce.
\newblock Reasoning with negative information, {II}: Hard negation, strong
  negation and logic programs.
\newblock In H.~{Pearce, D.; Wansing}, editor, {\em Proceedings of the
  International Workshop on Nonclassical Logics and Information Processing},
  volume 619 of {\em LNAI}, pages 63--79, Berlin, FRG, November 1990. Springer
  Verlag.

\bibitem[Pfe89]{Pfenning89lics}
Frank Pfenning.
\newblock {Elf}: A language for logic definition and verified meta-programming.
\newblock In {\em Fourth Annual Symposium on Logic in Computer Science}, pages
  313--322, Pacific Grove, California, June 1989. IEEE Computer Society Press.

\bibitem[Pfe91]{Pfenning91lf}
Frank Pfenning.
\newblock Logic programming in the {LF} logical framework.
\newblock In G\'{e}rard Huet and Gordon Plotkin, editors, {\em Logical
  Frameworks}, pages 149--181. Cambridge University Press, 1991.

\bibitem[Pfe92]{Pfenning94un}
Frank Pfenning.
\newblock Computation and deduction.
\newblock Unpublished lecture notes, 277 pp. Revised May 1994, April 1996, May
  1992.

\bibitem[Rei78]{Reiter78}
R.~Reiter.
\newblock On closed world databases.
\newblock In Gallaire and Minker, editors, {\em Logic and Databases}, pages
  55--76. Plenum Press, New York, 1978.

\bibitem[RP96]{Rohwedder96esop}
Ekkehard Rohwedder and Frank Pfenning.
\newblock Mode and termination checking for higher-order logic programs.
\newblock In Hanne~Riis Nielson, editor, {\em Proceedings of the European
  Symposium on Programming}, pages 296--310, Link{\"o}ping, Sweden, April 1996.
  Springer-Verlag LNCS 1058.

\bibitem[SH93]{SchroederHeister93lics}
Peter Schroeder-Heister.
\newblock Rules of definitional reflection.
\newblock In M.~Vardi, editor, {\em Proceedings of the Eighth Annual {IEEE}
  Symposium on Logic in Computer Science}, pages 222--232, Montreal, Canada,
  June 1993.

\bibitem[SP98]{Schurmann98cade}
Carsten Sch{\"u}rmann and Frank Pfenning.
\newblock Automated theorem proving in a simple meta-logic for {LF}.
\newblock In Claude Kirchner and H{\'e}l{\`e}ne Kirchner, editors, {\em
  Proceedings of the 15th International Conference on Automated Deduction
  (CADE-15)}, pages 286--300, Lindau, Germany, July 1998. Springer-Verlag LNCS
  1421.

\bibitem[Vir99]{Virga99phd}
Roberto Virga.
\newblock {\em Higher-Order Rewriting with Dependent Types}.
\newblock PhD thesis, Department of Mathematical Sciences, Carnegie Mellon
  University, 1999.
\newblock Forthcoming.

\bibitem[Wan93]{Wansing93}
Heinrich Wansing.
\newblock {\em The logic of information structures}, volume 681 of {\em Lecture
  Notes in Artificial Intelligence and Lecture Notes in Computer Science}.
\newblock Springer-Verlag Inc., New York, NY, USA, 1993.
\newblock Revision of the author's doctoral thesis (Fachbereich Philosophie und
  Sozialwissenschaften I of the Free University of Berlin).

\end{thebibliography}

\section{Refinements}
\label{sec:ref}

In the following sections we will move towards an operational
semantics for our language.


% \subsection{The Algebra of Clause Complementation}
% \label{ssec:ccalgebra}

% Similarly to what we did for term complementation, we show that we can
% organize  programs in a boolean algebra under $\la
% \And,\tensor,\mnotd,\top,\perp\ra$, generalizing the approach in \cite{MP90}.
% \bmeta fill in \emeta

\subsection{More on Termination}
\label{ssec:ccterm}

We prove that clause complementation preserve
termination, so that a negative program can be used as decision
procedure if so is its positive counterpart. First some
preliminaries: call $[R]$ the stable and transitive closure of the
subgoal ordering defined in Figure \ref{fig:termgraph}. Define
$\mnot(R)$ as follows:
\beq
\mnot(\cdot) & = & \cdot\\
\mnot(R,G_1\less G_2) & = & \mnot(R),\mnotg(G_1)\less\mnotg(G_2)
\eneq
Observe that, by the restriction to clauses of the form $Q\If G$, it
is sufficient to consider only goal complementation. It is clear that
negation is preserved under relation union, that is:
\beq
\mnot(R_1,R_2) & = & \mnot(R_1),\mnot(R_2)
\eneq
We will also need the following technical Lemma, stating that goal
complementation commutes with substitution:
\begin{Lemma}
\label{le:notgtheta}
For any substitution $\q$, $\mnotg([\q] G) = [\q] \mnotg(G)$.
\end{Lemma}
\begin{proof}
  By a simple induction on the structure of $\mnotg(G)$.
\end{proof}

\newcommand{\qq}{[\theta]}
%Next we show that
\begin{Lemma}
\label{le:notwf}
If $[R]$ is well-founded, so is $[\mnot(R)]$.
\end{Lemma}
\begin{proof}
  Suppose $[\mnot(R])$ is not well-founded. Then there is an infinite
  descending chain $\dots\less\qq G_i\less\dots\less\qq G_2\less\qq
  G_1$. By definition of $\mnot(R)$ there are $G'_i$ such that
  $\mnotg(G'_i)=G_i$ and that $G'_i\less G'_i-1\in R$ for
  $i\in\omega$. Since by Lemma \ref{le:notgtheta} $\mnotg([\q] G_i) =
  [\q] \mnotg(G_i)$, then $[R]$ allows an infinite descending chain
  $\dots\less\qq G'_i\less\dots\less\qq G'_2\less\qq G'_1$,
  impossible.
\end{proof}

The final piece is to show that the relation induced by the complement
of a program is the complement of the relation induced by the program
itself:
\begin{Lemma}
\label{le:ccterm}
\mbox{}
\ben
\item If $\pi::D\yd R$ then $\mnotd(D)\yd\mnot(R)$.
\item If $\gamma::\GG\vd G\yg R$ then $\GG\vd\mnotg(G)\yg\mnot(R)$.
\enen
\end{Lemma}
\begin{proof}
By induction on the structure of the given derivations. We prove some
selected cases:
\bit
\itc $\pi::\top\yd \cdot$; $\mnotd(\top)=\perp\yd\cdot=\mnot(\cdot)$.
\itc $\pi$ ends in $(q\ \vM\If G)\yd R,G\less q\ \vM$:
\btab
$\cdot\vd G\yg R$\bs
$\cdot\vd\mnotg(G)\yg\mnot(R)$\bi $2$\\
$\BAnd_{\vN\in\mnot(\vM)}\Pi( \neg (q\ \vN)) \And  \neg q\ \vM\If\mnotg(
G)\yd\mnot(R),\mnot(G)\less\neg q\ \vM$\\
\br\\
$\mnotd( q\ \vM\If G) \yd\mnot(R,G\less q\ \vM)$\br
\etab
\itc $\gamma::Q\yg \cdot$; $\mnotd(Q)=\neg Q\yd\cdot=\mnot(\cdot)$.
\itc $\gamma$ ends in $\GG\vd G_1\And G_2\yg R_1, R_2,G_1\less G_1\And G_2,G_2\less G_1\And G_2$:
\btab
$\GG\vd G_1 \yg R_1$ and $\GG\vd G_2\yg R_2$\bs
$\GG\vd \mnotg(G_1) \yg \mnot(R_1)$ and $\GG\vd \mnotg(G_2)\yg \mnotg(R_2)$\bi
$2$\\
$\GG\vd \mnot(G_1)\Or \mnotg(G_2)\yg \mnot( R_1), \mnot(R_2)$,\\
$\qquad\mnotg(G_1)\less \mnotg(G_1)\Or
\mnotg(G_2),\mnotg(G_2)\less \mnotg(G_1)\Or \mnotg(G_2)$\br\\
$\GG\vd\mnot(G_1\And G_2)\yg \mnot( R_1), \mnot(R_2),$\\
$\qquad\mnotg(G_1)\less \mnotg(G_1\And
G_2),\mnotg(G_2)\less \mnot(G_1\And G_2)$\br\\
$\GG\vd\mnot(G_1\And G_2)\yg \mnot( R_1, R_2,G_1\less G_1\And
G_2,G_2\less G_1\And G_2)$\br
\etab
\enit
\end{proof}

\begin{Corollary}
  If $\cP$ is terminating, so is $\mnotd(\cP)$.
\end{Corollary}
\begin{proof}
  By definition, $\cP$ is terminating if $\cP\yd R$ and $[R]$ is
  well-founded: by Lemma \ref{le:ccterm} $\mnotd(\cP)\yd\mnot(R)$ and
  by Lemma \ref{le:notwf} $[\mnot(R)]$ is well-founded. Thus
  $\mnotd(\cP)$ is terminating.
\end{proof}


\subsection{Elimination of $\tensor$}
\renewcommand{\vN}{\vec{N}}
\label{ssec:tensor}

We now show how to eliminate the $\tensor$ operator preserving 
provability; this will recover uniformity in proof-search.  
We generalize the approach in 
\cite{Bar90}, where the operation was defined as follows:
for $m1: Q_{1}\If G_{1}$ and  $m2: Q_{2}\If G_{2}$:
\beq
m1 \tensor m2 & = &  \q(Q_{1}\If G_1\And G_2) \quad\mbox{{\rm where }} 
\q = mgu(Q_{1},Q_{2}) 
\eneq
As we remarked earlier, the $\tensor$ operator was introduced to
preserve the duality between conjunction and disjunction in clauses.
Its use is limited to clauses in the same predicate definition and
therefore it can be eliminated simulating unification in the
definition.  Yet, the strict \ho unification problem is quite complex
and even more so complicated by the mixed quantifier structure of HHF.
Moreover, we have already in our language parameter (dis)equations
introduced by the augmentation procedure, as well as variable-variable
(dis)equations stemming from left-linearization.  We thus choose to
compile our source into an intermediate language which makes
unification problems explicit as simple equational problems in the
style of the unification logic introduced in \cite{Pfenning91lf}.  We
can then perform constraint simplification
as used in \emph{Elf} \cite{Pfenning89lics} and \emph{Twelf}
\cite{Schurmann98cade}.

% \newcommand{\eprv}{\stackrel{\prv}{\sim}}
% \newcommand{\enprv}{\stackrel{\nprv}{\sim}}
% \smallskip

% We introduce a notion of \emph{operational equivalence} \wrt 
% provability.

% \begin{Definition}[Operational Equivalence]
% \label{def:opeq}
% We say that $\cP_{1}$ and $\cP_{2}$ are \emph{operationally equivalent \wrt 
% provability}, denoted $\cP_{1}\eprv \cP_{2}$, iff, for any goal $G$ we 
% have $\ga\prv_{\cP_{1}} G$ iff $\ga\prv_{\cP_{2}} G$.

% \end{Definition}

We adapt the \emph{residuation} technique used in \cite{Pfenning94un}
to compile immediate implication into resolution.  We define the
judgment $D_{1}\tensor D_{2} \setminus D$ in Figure \ref{fig:otimes}
by induction on (normalized) $D_{1}$, with the intended meaning of
`$D_{1}\tensor D_{2}$ compiles to $D$': As usual, we add the rules
symmetric to $\tensor\top,\tensor\Pi,\tensor\And$.  Note that
residuation is undefined for clauses with different heads; yet we
could make the judgment total by agreeing to compile them to $\top$.

% \begin{Lemma}
% \label{le:imm-eq}
% The following are valid rules of inference:
% $$\ianc{\ga\sp q\ \vN_{1}\imm Q\quad \vN_{1}\eqd \vN_{2}}{\ga\sp q\ \vN_{2}\imm Q}{\imm\eqd}
% \qquad
% \ianc{\gna\fp q\ \vN_{1}\nimm Q\quad \vN_{1}\neqd \vN_{2}}{\ga\fp q\ \vN_{2}\nimm Q}{\imm\neqd}
% $$
% \end{Lemma}
% \begin{proof}
% It will be trivial once we specify the equality rules.
% \end{proof}

\begin{figure}
\[
\begin{array}{c}
\ianc{}{\top\tensor D \setminus \top}{\tensor\top}
\vsk
\ianc{}{(q\ \vN_{1}\If G_{1})\tensor  (q\ \vN_{2}\If G_{2}) \setminus
q\ \vN_{1}\If (\vN_{1}\eqd \vN_{2}) \And G_{1}\And G_{2}}{\tensor{\rm 
atm}}\vsk
\qquad
\ianc{([u/x] D_{1})\tensor D_{2}\setminus[u/x]D}{(\PIXA D_{1})\tensor 
D_{2}\setminus \PIXA  D}{\tensor\Pi}\vsk
\ianc{D_{1}\tensor D_{2}\setminus D\quad D'_{1}\tensor D_{2}\setminus D'}
{(D_{1}\And D'_{1})\tensor D_{2}\setminus D\And D'}{\tensor\And}
\end{array}
\]
\caption{$\tensor$-Elimination: $D_{1}\tensor D_{2}\setminus D$}
\label{fig:otimes}
\end{figure}

\begin{Example}
\label{ex:closedatensor}
Continuing Example \ref{ex:closed},
recall that:
\beq
\lefteqn{\mnotd(cloz)=}\\
& & \PI{F}{exp}\neg closed\ (lam\ F)\And \PI{F_1,F_2}{exp}
\neg closed\ (app\ F_{1} \ F_{2}).\\
\lefteqn{\mnotd(clolam)=}\\
&& \neg closed\ z\And  \PI{F_1,F_2}{exp}\neg closed\
               (app\ F_{1} \ F_{2})\And  \\ 
&& \PI{E}{exp} \neg closed\ (lam\ E) \If \PI {x } {exp}\ (closed\ x \Imp \neg closed\ 
 (E\ x)).
\eneq
Let us compute $\mnotd (cloz) \otimes \mnotd (clolam)\setminus D$:
\[\begin{array}{l}
 \PI{F}{exp}\neg closed\ (lam\ F)\If (lam\ F)\eqd z\And\mbox{}\\
 \PI{F,F_1,F_2}{exp}\neg closed\ (lam\ F)\If (lam\ F)\eqd (app\ F_1\
 F_2)\And\mbox{}\\
\PI{E}{exp}\neg closed\ (lam\ E) \If (\PI {x } {exp}\ closed\ x \Imp
 \neg closed\ (E\ x))\And\mbox{}\\
 \PI{F_1,F_2}{exp} \neg closed\ (app\ F_{1} \ F_{2})\If (app\ F_{1} \
 F_{2})\eqd z\And\mbox{}\\
 \PI{F,F_1,F_2}{exp} \neg closed\ (app\ F_{1} \ F_{2})\If (app\ F_{1} \
 F_{2})\eqd (lam\ F)\And\mbox{}\\
 \PI{F_1,F_2}{exp} \neg closed\ (app\ F_{1} \ F_{2}).
\end{array}\]

Note that only the third and last clause are satisfiable.
\end{Example}

The following lemma guarantees that compilation preserves immediate 
entailment:

\begin{Lemma}
\label{le:te-tim}
Let $ D_{1}\tensor D_{2}\setminus D$:  $\ga\vd D_1\tensor D_{2}
\imm Q$ iff $\ga\vd D\imm Q$.
\end{Lemma}
\begin{proof}
($\Imp$). By induction on the structure of $\pi:: D_{1}\tensor D_{2}\setminus D$.
\bit
\itc $\pi$ ends in $\setminus\top$. Trivially true.
\itc
$$\pi=\ianc{}{(q\ \vN_{1}\If G_{1})\tensor  (q\ \vN_{2}\If G_{2}) \setminus
q\ \vN_{1}\If (\vN_{1}\eqd \vN_{2}) \And G_{1}\And G_{2}}{\tensor{\rm 
atm}}$$
\btab
$\ga\sp (q\ \vN_{1}\If 
G_{1})\tensor (q\ \vN_{2}\If G_{2})\imm q\ \vN$\bh
$\ga\sp q\ \vN_{1}\If 
G_{1}\imm q\ \vN$ and $ q\ \vN_{2}\If G_{2}\imm q\ \vN$\bv
$\ga\sp\vN_1\eqd \vN$ and $\ga\sp G_1$\bv
 $\ga\sp\vN_2\eqd\vN$ and $\ga\sp G_2$\bv
$\ga\sp \vN_1\eqd\vN_2$\`By replacement\\
 $\ga\sp q\ \vN_{1}\If (\vN_{1}\eqd \vN_{2}) \And G_{1}\And 
G_{2}\imm q\ \vN$\br(s)
\etab
\itc $\pi$ ends in $\tensor\And,\tensor\Pi$: by an immediate
appeal to the IH.
% Case: 
% $$\ianc{([t/x] D_{1})\tensor D_{2}\setminus [t/x]D}{(\PIXA D_{1})\tensor 
% D_{2}\setminus \PIXA  D}\tensor\Pi}$$
% By inversion $([t/x] D_{1})\tensor D_{2} \setminus[t/x]D$ and $\ga\sp 
% [t/x]D\imm Q$ which is equivalent by IH to $\ga\sp [t/x]D_{1}\tensor 
% D_{2}\imm Q$ iff $\ga\sp [t/x]D_{1}\imm Q$ and $\ga\sp D_{2}\imm Q$ 
% iff $\ga\sp (\PIXA D_{1})\tensor D_{2}\imm Q$.
\enit
($\If$): completely analogous.
\end{proof}
We remark that, using again terminating programs, the above Lemma
covers immediate denial as well.
% The following guarantees that compilation preserves 
% Note that we have to take into account the meta-level universal 
% quantification on terms brought in by the $I^{\nprv}_{\Pi}$ rule.

% \begin{Lemma}
% \label{le:te-nimm}
% For all $\q, \ \q D_{1}\tensor D_{2}\setminus D$ and $\q D\nimm Q$ iff 
% $\q D_{1}\tensor D_{2} \nimm Q$.  
% \end{Lemma}
% \begin{proof}
% By induction on $D_{1}$.  We avoid mentioning substitutions when they 
% are immaterial and show only some significant cases in the $\Imp$ 
% direction, the converse following the same lines.
% \bit
% \item
% Case:
% $$\ianc{}{\top\tensor D \setminus \top}\tensor\top}$$
% $\ga\sp\top\nimm Q$ obviously entails   $\ga\sp\top\tensor D\nimm Q$.
% \item
% Case:
% $$\ianc{}{(q\ \vN_{1}\If G_{1})\tensor  (q\ \vN_{2}\If G_{2}) \setminus
% q\ \vN_{1}\If \vN_{1}\eqd \vN_{2} \And G_{1}\And G_{2}}\tensor{\rm 
% atm}}$$
% Assume $\ga\fp q\ \vN_{1}\If \vN_{1}\eqd \vN_{2} \And G_{1}\And 
% G_{2}\nimm Q$.  Then either $\ga\fp q\ \vN_{1}\nimm Q$ or $\ga\fp 
% \vN_{1}\eqd \vN_{2} \And G_{1}\And G_{2}$. The first case entails
% $\ga\fp q\ \vN_{1}\nimm Q$ or $\ga\fp G_{1}$ which implies 
% $\ga\fp q\ \vN_{1}\If G_{1}\nimm Q$ and finally 
% $\ga\fp q\ \vN_{1}\If G_{1}\tensor q\ \vN_{2}\If G_{2}\nimm Q$. The 
% second horn splits into three subcases:
% \ben
% \item
% $\vN_{1}\neqd \vN_{2}$. By Lemma \ref{le:imm-eq} $\ga\fp q\ 
% \vN_{2}\nimm Q$ which implies $\ga\fp q\ \vN_{2}\If G_{2}\nimm Q$ and finally 
% $\ga\fp q\ \vN_{1}\If G_{1}\tensor q\ \vN_{2}\If G_{2}\nimm Q$. 
% \item
% $\ga\fp G_{2}$ implies $\ga\fp q\ \vN_{2}\If G_{2}\nimm Q$ and finally 
% $\ga\fp q\ \vN_{1}\If G_{1}\tensor q\ \vN_{2}\If G_{2}\nimm Q$. 
% \item
% $\ga\fp G_{1}$ implies $\ga\fp q\ \vN_{1}\If G_{1}\nimm Q$ and finally 
% $\ga\fp q\ \vN_{1}\If G_{1}\tensor q\ \vN_{2}\If G_{2}\nimm Q$. 
% \enen
% \item
% Case:
% $$\ianc{([t/x] D_{1})\tensor D_{2}\setminus [t/x] D}{(\PIXA D_{1})\tensor 
% D_{2}\setminus \PIXA  D}\tensor\Pi}$$
% From $([t/x] D_{1})\tensor D_{2}\setminus [t/x] D$ and $\ga\sp \q 
% D\nimm Q$ by IH we infer $\ga\sp \q D\tensor D_{2}\nimm Q$, 
% equivalent to $\ga\sp \q D\nimm Q$ or $ D_{2}\nimm Q$ which implies 
% $(\PIXA D_{1})\tensor D_{2}\nimm Q$. 
% \enit
% \end{proof}
\newcommand{\ted}[1]{ (#1)^{\tensor_D} }
\newcommand{\teg}[1]{ (#1)^{\tensor_G} }
\newcommand{\PT}{\tensor\cP}

We now extend the effect of the above compilation on a program:
\begin{Definition}
\label{def:tensorpro}
\[\begin{array}{rcll}
 \ted{\top} & = & \top\\
\ted{\perp} & = & \perp\\
\ted{Q\If G} & = & Q\If \teg{G}\\
\ted{ D_{1}\And D_{2} } & = & \ted{D_{1}}\And \ted{D_{2}} \\
\ted{\PI{x}{A} D} & = & \PIXA  \ted{D} \\
\ted{D_{1}\tensor D_{2}} & = &  D  
& \mbox{\textrm{where }} \ted{D_{1}}\tensor \ted{D_{2}} \setminus D\\
\teg{X} & = & X & X\in\{\top,\perp,Q,\mbox{(dis)eq}\}\\
\teg{G_1\And G_2} & = & \teg{G_1}\And \teg{G_2}\\
\teg{G_1\Or G_2} & = & \teg{G_1}\Or \teg{G_2}\\
\teg{D\Imp G} & = & \ted{D}\Imp\teg{G}\\
\teg{\PIXA G} & = & \PIXA\teg{G}\\
\teg{\exa G} & = & \exa\teg{G}
\end{array}\]
\end{Definition}


\begin{Theorem}[Elimination of $\tensor$]
\label{thm:te-imm}
\mbox{} 
\ben
\item
 $\cS :: \ga\sp G$ iff $\tensor \cS :: \GG;\ted{\cD}\sp\teg{ G}$;
\item
 ${\cal I} :: \ga\sp D \imm Q$ iff $\tensor {\cal I} :: \GG;\ted{\cD}\sp
\ted{D} \imm Q$.
\enen
\end{Theorem}
\begin{proof}
($\Imp$). By a straightforward mutual induction on the structure of the given
derivations: we show the crucial case.
\bit
\itc $\cal I$ ends in $\imm\tensor$:
\btab
$\ga\sp D_1\imm Q$ and $\ga\sp D_2\imm Q$\bs
$\GG;\ted{\cD}\sp \ted{D_1}\imm Q$ and $\GG;\ted{\cD}\sp \ted{D_2}\imm
Q$\bi $2$\\
$\GG;\ted{\cD}\sp\ted{D_1}\tensor \ted{D_2}\imm Q$\br $\imm\tensor$\\
$\GG;\ted{\cD}\sp\ted{D_1\tensor D_2}\imm Q$\` By Lemma \ref{le:te-tim}
\etab
\enit
($\If$). Similarly.
\end{proof}
The same remark \wrt termination applies.
%\begin{Proposition}
% \label{ppn:te-nimm}
% $\cP \enprv \PT$.
% \end{Proposition}
% \begin{proof}
% We need to show $\ga\nprv_{\cP} G$ iff $\ga\nprv_{\PT} G$. We 
% generalize this to:
% \ben
% \item
% $\SM :: \ga\fp G$ iff $\tensor \SM :: \ga\nprv_{\PT} G$;
% \item
% For every $\q$, $\IM :: \ga\fp \q(def(Q)) \nimm Q$ iff $\tensor \IM :: \ga\nprv_{\PT} 
% \q(def(Q)) \nimm Q$.
% \enen
% The proof is a straightforward mutual induction on the structure of the 
% derivation using Lemma \ref{le:te-nimm}.
% \end{proof}

% From the above propositions, we have the soundness and completeness 
% of the compilation:

% \begin{Corollary}
% \label{co:te-ti}
% $\cP \sim \PT$.
% \end{Corollary}

\subsection{Constraint Simplification}
\label{sec:simplify}

In this section we hint on how to bring clauses stemming from
complementation into their final format by simplifying, whenever
possible, the disequation introduced by the $\mnota\Imp$ rule. In our
language, two forms of disequations cannot be immediately solved and
must be postponed as constraints:
\begin{enumerate}
\item Shallow parameter expression/existential variable disequations.
\item Rigid term/existential variable disequations, stemming from
  left-linearization.
\end{enumerate}

What can be solved instead is:
\begin{itemize}
\item The rest of existential variable-rigid term disequations by
  computing the complement of the latter.
\item Unsatisfiable variable-variable disequations, by removal.
%\item Removing useless augmentation clauses brought in by subordination.
\end{itemize}

In a final pass we may also:
\begin{itemize}
\item Remove irrelevant occurrences of $\top$.
%\item Remove failing clauses, \ie with $\perp$ in the body.
\item Hide the irrelevant augmentation in the final negative clause.
\item If so desired and when possible, bring clauses to the `core'
  languages syntax, as in LF.
\end{itemize}



\begin{Example}
\label{ex:closedfinal} 
Continuing Example \ref{ex:closed} and \ref{ex:closedatensor} the final
definition of $\neg$\texttt{closed} is:
\beq
oplam &: & \PI{ E}{exp} open\ (lam\ E)  \If \PI {x } {exp}\ open\ (E\ x).\\
opap1 &: & \PI{ E_{1},E_{2}}{exp} open\ (app\ E_{1} \ E_{2}) \If open\
E_1.\\
opap2 &: & \PI{ E_{1},E_{2}}{exp} open\ (app\ E_{1} \ E_{2}) \If open\ E_{2}.
\eneq
where we have solved all disequations, which are in this case trivial
ground-ground checks,  hidden the irrelevant assumption $closed\ x$ and
renamed `$\neg$\texttt{closed}' into \texttt{open}.
\end{Example}

\begin{Example}
\label{ex:linearfinal} 
The final definition of \texttt{$\neg$linear} and in turn \texttt{$\neg$linx} is:
\beq
\neg linapp1 & : & \neg linear(app\ E_1\ E_2)\If \neg linear(E_1).\\
\neg linapp2 & : & \neg linear(app\ E_1\ E_2)\If  \neg linear(E_2).\\
\neg linlam1& : & \neg linear(lam\ \lx E\ x)\If \neg linx( \lx E\ x)\\.
\neg linlam2& : & \neg linear(lam\ \lx E\ x)\If (\PI{y}{exp} (\neg linx(\lx y)\And linear(y))\Imp\neg  linear(E\ y)).\\
% \neg linx1& :& \neg linx(\lx app\ (E_1\ x) \ (E_2\ x))\If strict(E_2\ x).\\
% \neg linx2& :& \neg linx(\lx app\ (E_1\ x) \ (E_2\ x))\If
% strict(E_1\ x).\\
\neg linxap0& :& \neg linx(\lx app\ (E_1\ x^1) \ (E_2\ x^1)).\\
\neg linxap1& : & \neg linx(\lx app\ (E_1\ x) \ (E_2\ x^1))\If \neg linx(\lx
 E_1\ x).\\ 
 \neg linxap2& : & \neg linx(\lx app\ (E_1\ x^1) \ (E_2\ x))\If \neg linx(\lx E_2\ x).\\ 
\neg linxap3& : & \neg linx(\lx app\ (E_1\ x) \ (E_2\ x))\If \neg \neg
linx(\lx E_1)\And \neg linx(\lx  E_2\ x).\\
\neg linxlm& : &\neg linx(\lx lam(\ly E\ x\ y))\If\\
& & \qquad(\PI{y}{exp} \neg linx(\lx y)\Imp \neg linx (\lx E\ x\ y)).
\eneq
\end{Example}
